Author's response to reviews

Title: How well do blood folate concentrations predict dietary folate intakes in a sample of Canadian lactating women exposed to high levels of folate? An observational study.

Authors:

Lisa A Houghton (lisa.houghton@acadiau.ca)
Kelly L Sherwood (kellysherwood@yahoo.ca)
Deborah L O'Connor (deborah_l.oconnor@sickkids.ca)

Version: 5 Date: 20 September 2007

Author's response to reviews:


Dr. Lolu da-Silva
Assistant Editor, BMC-series journals

Dear Dr. da-Silva:

We are delighted to learn that our manuscript entitled, ¿How well do blood folate concentrations predict dietary folate intakes in a sample of Canadian lactating women exposed to high levels of folate? An observational study.¿ (MS: 1947964852129440) has been accepted for publication. We made the final few revisions suggested by the reviewers. We specifically used the wording provided by Dr. Jacques to bring our text in line with the statistical findings illustrated in Figure 1.

Do not hesitate to let us know if there are any further comments to be responded to in reference to this manuscript.

Sincerely,

Deborah L O'Connor PhD RD
Associate Professor, Department of Nutritional Sciences
University of Toronto
Director of Clinical Dietetics
Senior Associate Scientist
Program in Physiology and Experimental Medicine
The Hospital for Sick Children
555 University Avenue
Toronto, Ontario M5G 1X8